Author's response to reviews

Title: Primary myxofibrosarcoma of the parotid: case report

Authors:

Xu Li (wuyahaha@fmmu.edu.cn)
Xin Chen (chenxin6036@sina.com)
Zhao-hui Shi (shizhaohui1979@hotmail.com)
Yang Chen (chendoc@fmmu.edu.cn)
Jing Ye (yejing@fmmu.edu.cn)
Li Qiao (qiaoli@fmmu.edu.cn)
Jian-hua Qiu (qiujh@fmmu.edu.cn)

Version: 3 Date: 24 May 2010

Author's response to reviews:

Dear editor:
Thank you for your letter of May 13, 2010 regarding our manuscript entitled “Primary myxofibrosarcoma of the parotid: case report” (Ms. No.: 1285967114340179) that was submitted to BMC cancer for publication. We are very delighted to know that the case report has been accepted.

Please find our uploaded revision and response on separate sheets in which we have carefully addressed the issues raised in the letter, point by point as follow:

1. The consent in the manuscript has been written again to include the requested information.
2. Key words have been removed.
3. Initials for author names have been used in authors' contributions.
4. The manuscript has been checked again. The uploaded manuscript is the final revisions.

Thanks again for your acceptance of our manuscript for publication in your journal.

Sincerely,
Xu Li